Author's response to reviews

Title: Multiple component interventions for preventing falls and fall-related injuries among older people: systematic review and meta-analysis

Authors:

Victoria A Goodwin (v.goodwin@exeter.ac.uk)
Rebecca S Whear (R.S.Whear@exeter.ac.uk)
Rebecca A Abbott (R.A.Stanton@exeter.ac.uk)
Alison Bethel (A.Bethel@exeter.ac.uk)
Obioha C Ukoumunne (O.C.Ukoumunne@exeter.ac.uk)
Jo Thompson-Coon (J.Thompson-Coon@exeter.ac.uk)
Ken Stein (Ken.Stein@exeter.ac.uk)

Version: 2
Date: 22 October 2013

Author's response to reviews:

1. We have added a statement indicating that ethical approval was not required as this was a secondary study not involving human participants.
2. We have renamed the 'implications for practice' heading as 'conclusions'
3. We have also added figure legends to the main manuscript rather than the figure themselves